\begin{thebibliography}{999}

\bibitem[1]{Cierniak-offline-CPE-1997}
% article
M. Cierniak and W. Li.
\newblock Optimizing Java Bytecodes.
\newblock {\em Concurrency\hspace{.17em}---\hspace{.17em}Practice and
Experience}, 9(6):427--444, June 1997.

\bibitem[2]{Chambers-TR-Whole-Program-1996}
% techreport, manual, mastersthesis, or phdthesis
C. Chambers, J. Dean, and D. Grove.
\newblock Whole-Program Optimization of Object-Oriented Languages.
\newblock TR UW-CSE-96-06-02, Department of Computer Science and
Engineering, University of Washington, 1996.

\bibitem[3]{Cierniak-PACT-1997}
% booklet, unpublished, or misc
M. Cierniak and W. Li.
\newblock Interprocedural Array Remapping.
\newblock To appear in {\em {}Proceedings of the International Conference on
Parallel Architectures and Compilation Techniques (PACT '97)}, San Francisco,
CA, November 1997.

\bibitem[4]{Cierniak-PLDI-1995}
% inproceedings
M. Cierniak and W. Li.
\newblock Unifying Data and Control Transformations for Distributed
Shared-Memory Machines.
\newblock In {\em Proceedings of the SIGPLAN '95 Conference on Programming
Language Design and Implementation}, La Jolla, CA, June 1995.

\bibitem[5]{Cierniak-COMPCON-1997}
% inproceedings
M. Cierniak and W. Li.
\newblock Briki: an Optimizing Java Compiler.
\newblock In {\em Proceedings of the IEEE COMP{}CON '96}, San Jose, CA,
February 1997.

\bibitem[6]{Aho-compilers-1986}
% book or inbook
A.~V. Aho, R. Sethi, and J.~D. Ullman.
\newblock {\em Compilers: Principles, Techniques, and Tools}.
\newblock Addison-Wesley, Reading, MA, 1986.

\bibitem[7]{Wolfe-compilers-1996}
% book or inbook
M. Wolfe.
\newblock {\em High Performance Compilers for Parallel Computing}.
\newblock Addison-Wesley Publishing Company, Reading, MA, 1996.

\bibitem[8]{Kaffe-v0.8.3-1997}
% booklet, unpublished, or misc
T. Wilkinson.
\newblock Kaffe v0.8.3 --- A Free Virtual Machine to Run Java Code.
\newblock March 1997.
\newblock Available at {\tt http://www.kaffe.org/}.

\bibitem[9]{Java-Language-Specification-1996}
% book or inbook
J. Gosling, B. Joy, and G. Steele.
\newblock {\em The Java Language Specification}.
\newblock Addison-Wesley Longman, Inc., Reading, MA, 1996.

\bibitem[10]{Java-VM-1996}
% book or inbook
T. Lindholm and F. Yellin.
\newblock {\em The Java Virtual Machine Specification}.
\newblock Addison-Wesley Longman, Inc., Reading, MA, 1996.

\bibitem[11]{SPEC-CPU92-1996}
% booklet, unpublished, or misc
Standard Performance Evaluation Corporation.
\newblock SPEC CPU92 Benchmarks.
\newblock April 1996.
\newblock Available at {\tt http://open.specbench.org/osg/cpu92/}.

\end{thebibliography}

% $Id: intro.tex,v 1.10 1997/07/30 17:35:43 cierniak Exp $

\section{INTRODUCTION}
\label{sec:intro}


Compiling scientific applications requires sophisticated compiler
optimizations.  Those optimizations are commonly available in
commercial compilers for languages traditionally used for high
performance computing.  Those languages (the most successful ones are
Fortran and, to a smaller extend, C++) are compiled off-line into a
stand-alone binary.  This approach is different from the dominant
compilation model for Java in which the source program is translated
into machine-independent {\tt class} files which are then interpreted
or dynamically (just-in-time) compiled into the target machine
language.

Many of the existing high-performance optimizations cannot be easily
adapted to just-in-time (JIT) compilation.  The reasons include:
\begin{itemize}
\item Sophisticated optimization techniques consume a lot of hardware
  resources: CPU time and memory.
\item Often, a global view of the program is necessary ({\em
    whole-program
    optimization\/}~\cite{Chambers-TR-Whole-Program-1996,
    Cierniak-PACT-1997}).
\end{itemize}

In this paper we address the issue of performing optimizations which
are as good or almost as good as traditional optimizations while
running much faster and using less memory.  For our study we use an
optimizations technique called {\em data transformations}.  In
previous work we have shown that data transformations can be very
effective in compiling Fortran or C
programs~\cite{Cierniak-PLDI-1995}.  Recently we have adapted the
technique for Java bytecodes~\cite{Cierniak-offline-CPE-1997,
  Cierniak-COMPCON-1997}.  However, our previous experiments with
optimizing Java bytecodes have been performed with an off-line
compiler and the compiler algorithms were not as fast as a competitive
JIT compiler is expected to be.  To perform data transformations we
need more program structure than is directly available in the
bytecodes.  Our previous implementation was recovering full high-level
structure of the original Java program before applying any
optimizations.  This high-level view of the input bytecodes was
represented with an intermediate representation called {\em JavaIR}.
JavaIR is very high-level and so high-performance optimizations, like
data remapping, can be performed very efficiently.  Unfortunately, our
experience has shown that recovering full structure from a {\tt class}
file is very expensive.  In our implementation, the time needed to
convert bytecodes to JavaIR is an order of magnitude longer than the
time to perform the optimizations.  Since most the high-level
structure is not needed to perform data transformation, we have
decided to use a different approach in our JIT optimizer.

This paper shows how to perform the same optimizations as described
in~\cite{Cierniak-offline-CPE-1997, Cierniak-COMPCON-1997} while
recovering only as much structure as needed and using faster (although
not necessarily as accurate) analysis techniques than those
traditionally used in off-line compilers~\cite{Aho-compilers-1986,
  Wolfe-compilers-1996}.  We have implemented all the techniques
described in this paper in our experimental compiler {\em Briki}.
Briki uses Kaffe~\cite{Kaffe-v0.8.3-1997} as the JIT compiler and
implements the optimizations in a relatively machine-independent
manner so that it can run Java programs on multiple architectures
(such as i386 or sparc).

